Author's response to reviews

Title: Risk Factors for Recurrent Clostridium difficile Infection (CDI) Hospitalization Among Hospitalized Patients with an Initial CDI Episode: A Retrospective Cohort Study

Authors:

Marya D Zilberberg (MZilberb@schoolph.umass.edu)
Kimberly Reske (kreske@dom.wustl.edu)
Margaret Olsen (molsen@dom.wustl.edu)
Yan Yan (yyan@dom.wustl.edu)
Erik R Dubberke (edubberk@dom.wustl.edu)

Version: 2 Date: 10 February 2014

Author's response to reviews:

We have now added the following verbiage to the beginning of the Methods section on page 4:

"This study was approved by the Washington University Institutional Review Board, and its conduct was in compliance with the Helsinki Declaration."

Thank you
Marya Zilberberg